Covering Letter

Nov. 4, 2005

The Chief Editor
BMC Pregnancy and Child Birth

Re: Resubmission of Revised version of Research Paper “Prevalence of pre- and postpartum depression in Jamaican women”

Dear Sir,

Thank you very much for accepting our paper in principle. Please find enclosed the revised version as requested.

The changes made are as follows:
1. Table 1 Column 4 Row 3 #17 has been replaced by #18 and 23.29% changed to 24.66%. Also Column 4 Row 4 #8 has been replaced by #7 and 10.96% changed to 9.59%.
2. Table 2 title: the word ‘Clinical’ has been replaced by ‘Demographic’.
3. Figure 1: the word ‘T3’ has been changed to ‘FT3’ and the word ‘T4’ has been changed to ‘TT4’.
4. In the ‘Discussion’ paragraph 2 line 2 remove ‘partum’.
5. In ‘Results’, under ‘thyroid status and depression’ paragraph 1 line 14 the word ‘T3’ has been changed to ‘FT3’.

Thanking You,

Omkar Parshad
Professor of Endocrinology